Author's response to reviews

Title: Safety, tolerability, and impact on allergic inflammation of autologous E.coli autovaccine in the treatment of house dust mite asthma - a prospective open clinical trial

Authors:

Markus A Rose (Markus.Rose@kgu.de)
Bianca Weigand (Bianca.Weigand@kgu.de)
Ralf Schubert (Ralf.Schubert@kgu.de)
Johannes Schulze (Johannes.Schulze@kgu.de)
Stefan Zielen (Stefan.Zielen@kgu.de)

Version: 7 Date: 1 June 2011

Author's response to reviews: see over
Dear Dr. Talbot, dear Mrs. Aguera, dear Miss Puebla,

We are very happy about your final decision and highly appreciate the elegant solution. Included please find the revised final version of our manuscript.

The study has been registered at ClinicalTrials.gov, the registry and results database of federally and privately supported clinical trials conducted in the United States and around the world, organized by the US National Institute of Health and meeting all of the ICMJE requirements. The study number is indicated on the front page of our manuscript.

We also modified the conflict of interest section as requested.

We do appreciate all your efforts and hope that our revised version fulfils your expectations.

Kind regards,

Markus A. Rose, M.D., M.P.H.
ass. Professor of Paediatrics
Dept. of Paediatric Pulmonology and Allergology